Author's response to reviews

Title: A comparative study of food habits and body shape perception of university students in Japan and Korea

Authors:

Ruka Sakamaki (sakamaki@seinan-jo.ac.jp)
Rie Amamoto (tanaka_r@seinan-jo.ac.jp)
Yoshie Mochida (mochida@seinan-jo.ac.jp)

Version: 2 Date: 20 September 2005

Author’s response to reviews:

Dear nutrition journal of editorial team
As the data of Korean subjects has been misplaced, it has made a change to the original values.
Regards
Ruka Sakamaki